Author’s response to reviewers

Title: Clinical benefits of corticosteroid administration during adult cardiopulmonary resuscitation: a systemic review and meta-analysis

Manuscript ID: 66407
(Invited manuscript ID: 05526890)

Version: 2 Date: 28 July 2021

We would like to thank the editor and reviewers for their helpful suggestions. We have already revised the manuscript and hope that the revised manuscript has met the requirements for publication.
Editor’s comments

1 Scientific quality: The manuscript describes a Meta-Analysis of clinical benefits of corticosteroid administration during adult cardiopulmonary resuscitation. The topic is within the scope of the WJCCM. (1) Classification: Grade B; (2) Summary of the Peer-Review Report: This manuscript is significance and have potential clinical impact in the management of adult cardiopulmonary resuscitation; (3) Format: There are 6 figures and 2 tables; (4) References: A total of 29 references are cited, including 9 references published in the last 3 years; (5) Self-cited references: There are no self-cited references; and (6) References recommendations: The authors have the right to refuse to cite improper references recommended by the peer reviewer(s), especially references published by the peer reviewer(s) him/herself (themselves). If the authors find the peer reviewer(s) request for the authors to cite improper references published by him/herself (themselves), please send the peer reviewer’s ID number to editorialoffice@wjgnet.com. The Editorial Office will close and remove the peer reviewer from the F6Publishing system immediately.

2 Language evaluation: Classification: Grade A. A language editing certificate issued by Filipodia was provided.

3 Academic norms and rules: The authors provided the PRISMA 2009 Checklist, and Biostatistics Review Certificate. No academic misconduct was found in the Bing search.

4 Supplementary comments: This is an invited manuscript. No financial support was obtained for the study. The topic has not previously been published in the WJCCM.

5 Issues raised: (1) The authors did not provide original pictures. Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor.

Answer: We have submitted a separated PowerPoint file (Figure all-pwp.pptx) in this revision. We have edited all figures to ensure that the resolution of the picture is 300 pixels/cm.

6 Re-Review: Not required.

7 Recommendation: Conditional acceptance.